Author's response to reviews

Title: Psycho-social impact of visual impairment on health-related quality of life among nursing home residents

Authors:

Mahesh Kumar Dev B. Optom (maheshdev2002@gmail.com)
Nabin Paudel PhD student (paudel.nabin@gmail.com)
Niraj Dev Joshi B. Optom (niraidevioshi@gmail.com)
Dev Narayan Shah MS (dn_shah@hotmail.com)
Shishir Subba PhD (shishirsubba21@gmail.com)

Version: 5 Date: 25 July 2014

Author's response to reviews: see over
I am delighted to inform you that our Editor has recommended the acceptance of your manuscript for publication in BMC Health Services Research. Before we can proceed, however, we must ask for the following Editorial Requests to be addressed.

We are very glad to hear that the editor has recommended the acceptance of our manuscript for publication in BMC Health Services Research.

We have addressed all the comments in the manuscript as suggested by the editor.

***********************

Editorial Requests

1. Title page

Please ensure each authors email address is present on the title page.

We have added each authors email address on the title page in this revision.

2. Conclusions

Please remove the heading 'Recommendations' at the end of your manuscript so that this section is encompassed within the Conclusions.

We have removed the heading 'Recommendations' and encompassed within the conclusion in the manuscript.

3. Acknowledgements

In order to comply with our Instructions for Authors (http://www.biomedcentral.com/bmchealthservres/authors/instructions/researcharticle#formatting-acknowledgements) we would ask that you remove the 'Financial support' heading and replace this with 'Acknowledgements'.

We have changed this as suggested so that it conforms to the journal style.

4. Authors' Contributions

Please format this section according to our Instructions for Authors: http://www.biomedcentral.com/bmchealthservres/authors/instructions/researcharticle#formatting-contributions
This includes referring to each author by their initials and a full sentence and paragraph format.

We have formatted this section in the manuscript so as to conform to the journal style.

5. Competing interests

Please rename the section currently called 'Conflicts of Interest' according to our Instructions for Authors. Where no competing interests are declared we recommend the following statement, "The authors declare that they have no competing interests'.

We have changed this as suggested by the editor.

**************************